Author's response to reviews

Title: Newborn Infection Control and Care Initiative for Health Facilities to Accelerate Reduction of Newborn Mortality (NICCI): study protocol for a randomized controlled trial.

Authors:

Chivorn Var (chivorn@rhac.org.kh)
Alessandra N Bazzano (abazzano@tulane.edu)
James C Welty (jwelty@tulane.edu)
Sudesh K Srivastav (ssrivas@tulane.edu)
Navapol Iv Ek (inavapol@gmail.com)
Richard A Oberhelman (oberhel@tulane.edu)

Version: 5 Date: 20 May 2015

Author's response to reviews:

We have removed the map from the manuscript document and added figure title and legend. Also, we have removed the word cluster from the title of the manuscript as well as in the submission system.